Author’s response to reviews

Title: Minimally invasive endoscopic treatment for lumbar pyogenic spondylitis: a retrospective study in a tertiary referral center

Authors:

Shih-Chieh Yang (skyspine@gmail.com)
Tsai-Sheng Fu (fts111@adm.cgmh.org.tw)
Hung-Shu Chen (edaspine@gmail.com)
Yu-Hsien Kao (ed102523@edah.org.tw)
Shang-Won Yu (ed102270@edah.org.tw)
Yuan-Kun Tu (ed100130@edah.org.tw)

Version: 3 Date: 5 September 2013

Author’s response to reviews:

Dear Editors:

Enclosed is a further revision of our paper, “Minimally invasive endoscopic treatment for lumbar pyogenic spondylitis: a retrospective study in a tertiary referral center” in which the issues raised by the Editors have been addressed. As your requirement, we have removed the Age and Sex on table 1 in the revised manuscript. To clarify the meaning of this article, we have also revised the title to be “Minimally invasive endoscopic treatment for lumbar pyogenic spondylitis: a retrospective study in a tertiary referral center”. The changes are highlighted by bold and underline in the revised manuscript. If you have further requirements or suggestions, please feel free to contact us. We will do our best to revise our manuscript.

We hope that our manuscript will now be acceptable for the peer review process in the BMC Musculoskeletal Disorder.

Sincerely,

Shih-Chieh Yang, Tsai-Sheng Fu, Hung-Shu Chen, Yu-Hsien Kao, Shang-Won Yu, Yuan-Kun Tu

Corresponding Author:

Hung-Shu Chen

Department of Orthopedic Surgery and Anesthesiology

E-DA Hospital / I-Shou University

No.1, E-DA Road, Kaohsiung City 82445, Taiwan, R.O.C.
Tel: +886-7-615-0011 ext. 2972
Fax: +886-7-615-0913
E-mail: edaspine@gmail.com